Author's response to reviews

Title: Clinical symptoms and chemotherapy completion in elderly patients with newly diagnosed acute leukemia: a retrospective comparison study with a younger cohort

Authors:

Rong Hu (hurong1246@126.com)
Yong Wu (wuyong9195@126.com)
Xiaoying Jiang (vic_999@yeah.net)
Wenteng Zhang (w.t98@163.com)
Le Xu (xl_8332@163.com)

Version: 3 Date: 11 May 2011

Author's response to reviews:

May 10, 2011
MS: 8897457834217527
Clinical symptoms and chemotherapy completion in elderly patients with newly diagnosed acute leukemia: a retrospective comparison study with a younger cohort
Rong Hu, Yong Wu, Xiaoying Jiang, Wenteng Zhang and Le Xu

Response to Letter from Natasha Mellins-Cohen
on behalf of Prof Dirk Vordermark
Tel: +44 (0) 20 3192 2013
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Dear Prof Dirk Vordermark:

Thank you for the opportunity to make additional revisions for our manuscript MS: 8897457834217527 Clinical symptoms and chemotherapy completion in elderly patients with newly diagnosed acute leukemia: a retrospective comparison study with a younger cohort based on re-review. As you will note, we have changed “primarily” to “newly” in the title as suggested by Reviewer 2. We have addressed all outstanding concerns as requested by both reviewers. Please pass along our sincere appreciation for their efforts on our re-submission.

We are submitting the revised manuscript with point-by-point changes noted in the paper, and have highlighted (with 'tracked changes'/coloured/underlines/highlighted text) all changes made when revising the manuscript. The paper conforms to the journal style.
This will serve as the cover letter giving a point-by-point response to the concerns expressed by the reviewers. Our responses on a point-by-point basis are indicated by bullet points.

Thank you again for considering our paper for publication in BMC Cancer.

Sincerely,

Xiaoying Jiang, Professor

Nursing School of Fujian Medical University

#1 Xueyuan Road, Shangjie Town, Minhou County, Fuzhou 350108, Fujian, P.R. China